Author's response to reviews

Title: Cystatin C as a potential confounder of osteoprotegerin levels in healthy men. A cross-sectional, observational study.

Authors:

Eva Kulcsar-Jakab (kulcsar.jakab.eva@gmail.com)
Zsofia Petho (pethozsofi@hotmail.com)
Edit Kalina (ekalina@med.unideb.hu)
Roza Foldesi (vfrozika@gmail.com)
Adam Balogh (balogha@med.unideb.hu)
Peter Antal-Szalmas (antalszp@med.unideb.hu)
Harjit P Bhattoa (harjit@med.unideb.hu)

Version: 2 Date: 9 February 2015

Author's response to reviews:

Kindly note that, as requested, the following editorial changes have been made to the manuscript:

Line numbers and page numbers have been included.
The name of the ethic committee (Ethics committee of the University of Debrecen, Hungary) has been added to the text (see lines 103-105).